Author’s response to reviews

Title: Intraosseous lipoma of the third lumbar spine: a case report

Authors:

Chaipond Teekhasaenee (teekha_c@hotmail.com)
Koji Kita (kitatatakita@docomo.ne.jp)
Kenji Takegami (tkbanboo@gmail.com)
Eiji Kawakita (eijiikawa17@yahoo.co.jp)
Toshihiko Sakakibara (sakakitoshi@mac.com)
Yuichi Kasai (ykasai@clin.medic.mie-u.ac.jp)

Version: 7 Date: 6 January 2015

Author’s response to reviews: see over
Thank you very much for Editor's comment. I hope that I have answered your questions adequately.

1. The case still needs copyediting for English Grammar. For instance, some sentences are written in the present and past tense at the same time, i.e. "is" instead of "was" etc.

Authors checked English Grammar and corrected some words and sentences.

2. The conclusion is not really a conclusion and reiterating the introduction. What is the take home / away message here? The authors should revised their abstract and conclusion after the discussion.

Authors re-wrote the section of conclusion.